
6MSD Weekly Scheduling Meeting 

Wednesday @ 1:30, MCC Conference Room 

August 10, 2011 

 

1) Glance Back and 2-Week Look Ahead (W. Oren) ................................................ 10 mins 

2) PMI Report (P. Collins) ........................................................................................... 5 mins 

3) Critical Path Service Provider Reports .................................................................. 20 mins 
(a) Alignment (Curtis) 
(b) Magnet Refurbishment (Baggett) 
(c) Installation/Alignment (Wilson) 
(d) HPRF (Kimber) 

4) Hall Status Reports ............................................................................................... 10 mins 

5) Special Topics: None .............................................................................................. 0 mins 

6) Safety Report  (M. Logue) ....................................................................................... 5 mins 

7) Meeting Recap and Action Items (All) ..................................................................... 5 mins 



6MSD: 
Glance back & look ahead

Will Oren for Fulvia Pilat

6MSD Meeting
August 10, 2011



Progress Wed Aug 3 – Wed Aug 10

Weeks: 13 down, 14 to go

Highlights:
• Magnet rework, measurement and installation on target to complete next 

week
• C100‐ 1‐ Installed, hookups proceeding with some challenges
Areas of attention:
• Continue vigilance on safety 
• Continue vigilance on cryomodule schedule: 4/2K cooldown, acceptance 

tests, installation, test with beam
• CHL 2K system recovery from battery “meltdown”
• Close up the SL  end of the week



CCB 
• The following 6MSD CRs & NTFs have been submitted, reviewed, and approved:

– Relocate ITV2C23 to girder 2C24 (NTF) ‐. Scope includes relocation of 2C23 YAG viewer and camera to 2C24 girder 
adjacent to Hall B radiator target ladder upstream from tagger magnet.

– NE Stub Demolition ‐ Change Method (CR) ‐ This change will allow demolition of the NE stub roof and east wall prior 
to the completion of tunnel extension including waterproofing. The proposal is to build a watertight bulkhead wall in 
the existing NE stub in lieu of the dust partition. Then, the roof and east wall will be saw cut into 4‐6 pieces ranging 
in weight from 0.5 ton to 13 tons and removed by a crane. Neat plan which helps hold schedule

• Change Requests/Notes to File Upcoming/Placeholders:
– FML Stair Repair at the FEL ‐ Block removal for egress (CR/NTF) ‐ FML's current plan in the schedule is to repair the 

FEL stairs when the wiggler arrives Sept 9th and during the 2 week wiggler installation period (starting Sept 
12th). FML is working with FEL on availability of time to perform this work as it needs to be coordinated with FEL 
run prior to the wiggler arrival and during wiggler installation. Additional work by MSINST group is needed to 
remove and reinstall all of the blocks at the truck ramp roll up door to support wiggler installation. One week notice 
to secure crane for block removal is required. Discussions Continue

– Tunnel Penetrations Repair (CR) ‐ Captures additional detailed scope for the repair work on defined 
penetrations. A&E final design drawings for the repair work is being reviewed by FML 2 August 2011. CR will be 
updated and resubmitted for repair (prototype) for existing penetrations (to be defined based on accelerator lock up 
schedules).

– System Testing/Hot Checkout/Restart Operations Plan (CR/NTF) ‐M. Spata is leading the preparation and 
planning for overall testing, hot checkout and restoration of operations for the 6 GeV run. The plan will be 
presented and discussed end of August.

– Hall C Ion Chamber/Dump Work (CR) ‐ Hall C ion chamber and dump work being reviewed by engineering division 
and accelerator operations for scope. Draft plan led by A. Freyberger has been submitted. Meetings are ongoing



4 weeks look ahead

• C100 ‐ 2 installation after Labor Day
• Completion of magnet  rework/reinstallation ~August 19th

• CM commissioning begins August 12th – S. Linac locked up day/swing shifts 
starting Aug 13th  , CHL Status?????

• BSY Penetration construction continues
• ESR Cooldown ready for endstation support early Sept.
• RF installation & testing continues
• LCW upgrades – Filling, Pressure Tests, System Cleanup, Cut in of Load
• Peppo installation continues
• Halls – HD‐ice to Hall B, G2P continues

• Facilities: TEDF, repair FEL stairs ?,  maintenance – leak repairs in tunnel?



4 week look ahead – warm topics
Safety and job planning
Vigilance

Lockup for cryo‐module testing – CHL Recovery for 2K Ops
In progress – could be a real challenge

Integrated plan for testing/hot check‐out/back‐to‐ops
In progress

Hall C Dump Work Scope/Schedule
In progress

Civil work schedule
Rebecca continuously monitoring and working on this. Neat plan for 

demolition “approved”



PM&IP

6MSD 8/10/11 Weekly Progress Meeting
(8/05/11 Data)

Report #13



PM&IP
• Schedule Issues: Hall D Phase 3 Tunnel Extension is ~5 

Days behind schedule.

• Hall A Septum Platform Parts Issues…

• Resource issues:  None



PM&IP
• Other Schedule Information at 

https://www.jlab.org/6MSD_Integrated_Schedule/SCHEDULES/
– Four Week Look Ahead - Shows all activities that have started 

that are not complete and any activities due to start in the next four 
weeks

– Critical Path – Shows all MUST activities that have 2 days or less 
float.

– Schedule Variance - Shows all MUST activities that are not 
100% complete and that the Start/Finish dates have slipped by at 
least 5 days.
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Shutdown Work Summary 
Organization: Alignment 

Name: Chris Curtis 

Date: 10th  Aug, 2011 

Last Week 

Work That Went Better Than / As Expected: 

• Step 2B (final alignment) of east arc south of the gate completed. 

• Step 2A of west arc / arc 10 continuing. 

• South linac warm region girder alignment. 

• G2P FZ2 magnet alignment, and septum magnet fiducialization. 

Work That Went Slower Than Expected, Delays, and Issues: 

• South SL24 cryomodule alignment – roll issue. 

Ongoing Shutdown Work 

 Table 1: Next Week's Work ( green  = no concerns;  yellow  = some concerns;  red  = serious concerns)  
Project & Customer  Status  Completion Tasks Comments/Concerns

12 Gev Arcs  X      TBD  Alignment of re‐installed dipoles 
in west and east arcs. 

Restarted step 2A in west arc.

Arc 10 Magnets  X      Opportunistic Quad / dipole Step 2A alignment. Restarted step 2a
Hall A  ‐ G2P  X      TBD  Harp fiducialization & motion 

check, FZ2 Magnet Alignment 
Continuing as appropriate. 
FZ2 magnet this week. 

12 Gev Linacs   
 

X    Before testing  Cryomodule, warm region & DP 
alignment 

Final alignment in south linac zones 
21‐27.  
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Table 2: Other Upcoming Work ( green  = no concerns;  yellow  = some concerns;  red  = serious concerns) 
Project & Customer  Status  Completion Tasks Comments/Concerns

12 GeV Arcs  X      TBD  Dipole re‐alignment. Continuing  
on north part of west and east 
arcs for the next 2+ weeks. 

Ongoing 

PEPPo  X      October  Layout and alignment of PEPPo 
components 

Due to start at end of the month. 

Q‐Weak    X    ????  Target alignment  When? 

 

New Work Requests Requiring Schedule Changes: 
• None 

Permits Not Already in Hand (hot work, lifting, blind penetrations): 
• None 

Significant Problems Pending (delays, technical issues, resource issues, scheduling): 
• None 

Staffing Outlook: 
• OK 
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Shutdown Work Summary 
Organization: Magnet Measurement 

Name: Ken Baggett 

Date: August 10, 2011 

Last Week 

Work That Went Better Than Expected: 
• Magnet Refurbishment – Four Arc 5 magnets remain to be measured. 
• Measurement – WA completed.   East Arc Stack6 being measured.  
• Installation – pacing with measurement 

Work That Went Slower Than Expected, Delays, and Issues: 
• Found measurement error on four Arc 2 dipoles.   Pulled, re‐measured, re‐installed. 
  

Ongoing Shutdown Work 

 Table 1: Next Week's Work ( green  = no concerns;  yellow  = some concerns;  red  = serious concerns)  
Project & Customer  Status  Completion Tasks Comments/Concerns

12GeV  x      12‐Aug  Refurbish Stacks   Last 4 dipoles currently on 
refurbishment stands. 

12GeV  x      16‐Aug  Measure Stacks  26 wire measurements and 3 
detailed maps remaining. 

12GeV  x      17‐Aug  Re‐install Stacks  WA installation complete. 
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Table 2: Other Upcoming Work ( green  = no concerns;  yellow  = some concerns;  red  = serious concerns) 
Project & Customer  Status  Completion Tasks Comments/Concerns

Injector    x    August 31 Map 2 magnets/complete 
solenoid. 

1st solenoid measurements being 
completed today. 

 

New Work Requests Requiring Schedule Changes: 
• None 

Permits Not Already in Hand (hot work, lifting, blind penetrations): 
• None 

Significant Problems Pending (delays, technical issues, resource issues, scheduling): 
• None 

Staffing Outlook: 
• Does anyone need any good technicians? 
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Shutdown Work Summary 
Organization: MSINST 

Name: Neil Wilson 

Date: 8/10/11 

Last Week 

Work That Went Better Than Expected/Completed: 
• SL Warm Region Girder– SL24 Installed 
• Fabricated, Tested & Installed Manifolds for Klystron Racks & CPS’s ‐ SL25 
• Fabricated Spools & Modified Dummy CM for SLINAC. Leak Checked.  
• LCW Tightness Checks ‐ West Arc South Completed 
• SLINAC Beam Line Installation, Pump Down, Bake Out & Leak Check Under Way. 
• West Arc North Beam Line Installation Progressing. 
• Polymer Specimen Out Gassing Study for G2P – 6 samples – Super Harp Fork 

Work That Went Slower Than Expected, Delays, and Issues: 
•  

Ongoing Shutdown Work 

 Table 1: Next Week's Work ( green  = no concerns;  yellow  = some concerns;  red  = serious concerns)  
Project & Customer  Status  Completion Tasks Comments/Concerns

12GeV  X        Dipole Hook Up ‐ LCW & Vac. West Arc North
12GeV          Fabricating Headers for 

Klystron Racks & CPS’s ‐ SL23 
In Cue

12GeV  X        Arc 10 Dipole & Girder 
Installation 

On Hold
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Project & Customer  Status  Completion Tasks Comments/Concerns

12GeV  X        Tying up Loose Ends/Punch List 
Items 

Misc. Tasks 

12GeV  X        LCW Hose Hook Ups   West Arc North‐On Going
Hall D  X        LCW Piping Installation  On Going
12GeV  X      8/12/11  SLinac Beam Pipe (Dummy Cryo 

Modules) Installation & Warm 
Region Girder’s Pumped Down, 
Bake Out, Leak Checked 

On Going 

12GeV  X        NLINAC CPS Installation  Delivery Date ? 
Hall A G2P  X        Misc. Installation Support  LCW, Air Line, etc. 
Hall D  X        Photon Beam Pipe Monitoring  On Going 
6GeV  X        Swapping Out Vacuum Valves  On Going 
FEL  X        Vacuum Support Minimal

Table 2: Other Upcoming Work ( green  = no concerns;  yellow  = some concerns;  red  = serious concerns) 
Project & Customer  Status  Completion Tasks Comments/Concerns

12GeV  X        Fabricating Headers for Klystron 
Racks & CPS’s 

Manifolds on Site

6GeV  X        Leak Check VIP1L0450 Wave 
Guide 

Trouble Shooting

6GeV  X        Vacuum Valve Air Line PM Maintenance

6GeV – Hall A  X        Change Out Ion Pmp’s @ DP 
Station & Change Out Conical 
Spool  

G2P  X        Continuing with Beam Line 
Installation, Pump outs & Leak 
Checking 
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Project & Customer  Status  Completion Tasks Comments/Concerns

6GeV – Hall B  X        Change Out Spool @ Tagger 
Radiator with 4‐way Cross 

6GeV – Hall C  X        #2 Moeller Quad Work Support 10” Bellows Removal, Re‐install 
Pump Out & Leak Check 

6GeV – Hall C  

Q‐Weak 

X        BCM Work‐Modification of 
IBC3H04 

Swap out IPM3H04 & IBC3H04 with 
BCM’s 

6GeV  X        Vacuum Valve Limit Switch 
Replacement 

AIPBSY    X      Relocate ITV2C23 to 2C24  Need additional Info.
11GeV  X        12GeV RF Separator Test Cavity  Pump down for testing. Needs to be 

moved to Vac Lab 
AOPS/G2P  X        Vacuum test to determine viewer 

deflection. 
Awaiting spring change in viewer

New Work Requests Requiring Schedule Changes: 
• None ‐ Misc. Minor jobs only. Nothing requiring Schedule Changes at this time.  
• Hall C Dump He Tube, work planning started. Actual start date TBD. 

Permits Not Already in Hand (hot work, lifting, blind penetrations): 
• None 

Significant Problems Pending (delays, technical issues, resource issues, scheduling): 
• Only 2 working days until the South is Locked up for Cryo Module Commissioning!!! 
• 10 weeks until Beam Ops Start‐ Tunnel Locked up North & South!!! 
• Time flies when you are having fun. 

Staffing Outlook: 
• Continues to be fair with some clouds on the horizon. Work force reduction started. 2 week notices give to 4 
contractors, termination date 8/19.  

• Dipole Installation Crew Completion date is ~ 8/19. This crew then rolls back to the Mech Inst Group. 
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Shutdown Work Summary 
Organization:  HPRF 

Name:  Andrew Kimber 

Date:  10 August 2011 

Last Week 

Work That Went Better Than/As Expected: 
• Testing of 2L24 

o Calibration of readbacks (forward and reflected power) 
o Eliminating noise, EMI, signal jitter – working much better 

• Installation of 2L25 
o LCW manifolds complete, hose connections started 

• Installation of 2L23 
o General cable installation continues,  card cages installation complete, LCW parts being worked on 

• Waveguide installed in tunnel in 2L24, shorting plates back on for more testing 

Work That Went Slower Than Expected, Delays, and Issues: 
• Evaluating circulator performance, may need additional ‘tweaking’ for full power 

Ongoing Shutdown Work 

 Table 1: Next Week's Work ( green  = no concerns;  yellow  = some concerns;  red  = serious concerns)  
Project & Customer  Status  Completion Tasks Comments/Concerns

12GeV HPRF  x      before tunnel 
lockup 

Further testing of 2L24  We will test as long as we can, 
depending on cryo 

12GeV HPRF  x      8/12/11  2L25 installation continues  LCW hoses, solenoid cabling (in fab 
shop) Will apply power when this is 
complete  
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Project & Customer  Status  Completion Tasks Comments/Concerns

12GeV HPRF  x      Late Aug  2L23 installation continues  General cabling, waveguide, LLRF, 
LCW, klystrons… 

 

Table 2: Other Upcoming Work ( green  = no concerns;  yellow  = some concerns;  red  = serious concerns) 
Project & Customer  Status  Completion Tasks Comments/Concerns

12GeV HPRF  x      8/19/11 2L25 Test into shorts complete
12GeV HPRF  x      Wk of 15th Aug? Installation of CPS#6 in 1L22
12GeV HPRF  x      Sept 1st Transfer to main LCW headers Skid no longer needed…
         

New Work Requests Requiring Schedule Changes: 
• n/a 

Permits Not Already in Hand (hot work, lifting, blind penetrations): 
• n/a 

Significant Problems Pending (delays, technical issues, resource issues, scheduling): 
• Updates from last week: 

o Solenoid PS still late…   now exploring other vendors 
o 2 FA HOM filters due the end of July, vendor reports a further delay in manufacturing, has consistently 

failed to update delivery schedule, very frustrating… 

Staffing Outlook: 
• Good 
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2L25, as of 9th August 2011 
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Waveguide installed in 2L24 (with shorting plates) 
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Work Summary 
Organization: Hall A 

Name: Folts 

Date: 8/9/11 

Last Week 

Work That Went Better Than Expected: 

Work That Went Slower Than Expected, Delays, and Issues: 
• Target stand parts  
• Target refrigerator fabrication 

Ongoing Shutdown Work 

 Table 1: Next Week's Work ( green  = no concerns;  yellow  = some concerns;  red  = serious concerns)  
Project & Customer  Status  Completion Tasks Comments/Concerns

G2P Water  x      continuing  Water manifold work   
 Septum magnet        continuing     
G2P Target    x    continuing  Modifying Septum platform  Parts are an issue 
G2P Target    x    continuing   Building refrigerator  Delayed due to Q week priority 
G2P  x      continuing  Power supply installation  Final hookup 
Hall maintenance   x      continuing  General repair and maintenance   
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Shutdown Work Summary 
Organization: Hall B 

Name: Doug Tilles 

Date:  8/10/2011 

Last Week 

Region 3&2 Sector 3 Drift Chamber              100% 
Remove R3S3 Steel work                                 100% 
Start installation of R3S2 Steel work             100% 

Work That Went Slower Than Expected, Delays, and Issues: 
• None 

Ongoing Shutdown Work 

 Table 1: Next Week's Work ( green  = no concerns;  yellow  = some concerns;  red  = serious concerns)  
Project & Customer  Status  Completion Tasks Comments/Concerns

 
6 GeV repairs 
Setup and pull  
Region 3 Sector 2 
Drift chamber  
Cosmic test of Drift 
Chamber 
 

X 
 
 
 
 
 
X 

    Wed Morning 
Start Region 3 
Sector 2 Repairs 
of Region 3 and 
2 
 
 
After Repairs 
 

   



6/22  Page 2 of 2 
 

Project & Customer  Status  Completion Tasks Comments/Concerns

HDice Lab work 
HDice to Hall 
Electron Run Blower 
Fab Blower suction 
line  

X 
X 
 

    On Going Target
8/15 
Electron blower  
pump on order 

   

12GeV Drift 
chamber Region 1 
shipment. 
 

X      ESB Building  
Ready for     
shipment 
Complete 

   

Table 2: Other Upcoming Work ( green  = no concerns;  yellow  = some concerns;  red  = serious concerns) 
Project & Customer  Status  Completion Tasks Comments/Concerns

• Region 3 S2 
removal and 
repair 

x      8‐10‐2011

         
         

New Work Requests Requiring Schedule Changes: 
• HDice  Hall Alignment         Sep 4 

Permits Not Already in Hand (hot work, lifting, blind penetrations): 
• None needed       Hot work complete 

Significant Problems Pending (delays, technical issues, resource issues, scheduling): 
• None 

Staffing Outlook: 
• OK 
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Shutdown Work Summary 
Organization: Hall C 

Name:  W. Kellner 

Date: 08‐10‐2011 

Ongoing Shutdown Work 

 Table 1: This Week's Work ( green  = no concerns;  yellow  = some concerns;  red  = serious concerns)  
Project & Customer  Status  Completion Tasks Comments/Concerns

6Gev Q‐weak        20%  Leak testing new beam line 
viewer manifold  

Unscheduled  

Other        50% Hall cleanup/painting  Unscheduled
6Gev Q‐weak    X    continuing  Moller magnet repair  Challenging 

Table 2: Upcoming Work ( green  = no concerns;  yellow  = some concerns;  red  = serious concerns) 
Project & Customer  Status  Completion Tasks Comments/Concerns

6Gev SOS         On hold   Prep work for removal of the SOS 
detector hut lead shielding 
panels  

Waiting for Decontamination 
approval/plan  

6Gev Q‐weak        On hold Reinstall main detector when 
repaired  

Waiting for users to complete repair 

Other         Started Pressure and leak test new SHMS 
instrumentation manifolds 

Unscheduled,  TOSP signed
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August 2011 Events
Date Location Event
08/01/11 Test Lab 

parking lot 
Employee was closing the van ramp and turned his head to the right. 
After getting in the van to leave he felt pain in the right side of his 
neck and was diagnosed with neck strain  and there was no treatment 
needed at this time.  Employee has returned to work with no 
restrictions.

08/02/11 TEDF Subcontractor was pulling in a 2" screw pipe using two pipe wrenches. 
Holding the pipe with one wrench and pulling with the other, he felt a 
pop in his right shoulder and was diagnosed with right shoulder 
strain.  Subcontractor was evaluated and there was no treatment 
needed at this time. Subcontractor  returned to work with no 
restrictions.

08/02/11 South Linac
zone 24. 

Two employees called to report that while measuring voltage, they 
felt a tingle in his right index finger. The sensation was immediately 
resolved. Employees reported minor electrical exposure and no 
medical assistance was needed at the time of notification. 
Employees returned to work on regular duty. 

08/08/11 Building 87 As employee was carrying boxes into building, the door began to close 
and the door latch scratched his arm.  

2



Safety Incentives
• JLab Safety‐On‐The‐Spot : 

– Recognizes safe behavior observed by any supervisor/manager. 
– Awarded by the supervisor/manager and includes a gift of modest value (flashlight, cup, 

lanyard, etc.). 
– Call Tina at 7611

• JLab Safety Achievement Award: 
– Recognizes individual employees that have performed above and beyond normal job 

duties to ensure the safety of their coworkers, the public, or the environment. 
– Cash award, nominated and approved for nominally $50 (after taxes) and is presented 

by the AD or COO and is accompanied by a JSA Certificate of Appreciation.

• JLab Director’s Safety Award: 
– Recognizes employees that have gone significantly above and beyond their normal job 

duties in achieving a major safety related accomplishment. 
– Cash award, nominated and approved for nominally $100 (after taxes) and is presented 

by the AD, COO, Deputy Director, or Laboratory Director.

http://www.jlab.org/ehs/safety_award.html


